

August 31, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 4634-review.doc).

Title: Alvarado, Eskelinen, Ohhmann and RIPASA scores for diagnosis of acute appendicitis

Author: Hasan Erdem, MD; Suleyman Cetinkunar, MD; Koray Daş, MD; Enver Reyhan, MD; Cumhur Değer, MD; Mehmet Aziret, MD; Hilmi Bozkurt, MD; Selçuk Uzun, MD; Selim Sözen, MD; Oktay İrkörücü, MD

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 4634

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 The reviewer 00181197 suggest a major revision about our paper . Revision has been made according to the suggestions of the reviewer.

The corrections:

1. The paper was sent to www.aje.com for language editing and certified as grade A
2. The title was shortened and rearranged
3. All authors' affiliations were rearranged
4. Author contributions were written separately
5. Results section was widened
6. Core Tip was added
7. All reference numbers format were corrected
8. Heading in "Material and Methods", was revised
9. Comments were added at the end.
10. In reference section, PMID and DOI numbers were added. The section was revised as told and a few ones were added
11. Figure 1 will be sent you as a separate file in word format so it can be decomposable format

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Suleyman Cetinkunar



Adana NUMUNE EĞİTİM ve ARAŞTIRMA HASTANESİ, Genel Cerrahi Kliniği. Serinevler Mah., Ege Bağtur Bulvarı, 01240 Yüreğir, ADANA-TURKİYE
Tel & Fax : 90 322 355 01 44